Author's response to reviews

Title: Comparative microRNA profiling of sporadic and BRCA1 associated basal-like breast cancers

Authors:

Max Yan (max.yan@sesiahs.health.nsw.gov.au)
Kristy Shield-Artin (kristy.shield@bakeridi.edu.au)
David Byrne (david.byrne@petermac.org)
Siddhartha Deb (siddhartha.deb@petermac.org)
Nic Waddell (nick.waddell@imb.uq.edu.au)
kConFab Investigators (heather.thorne@petermac.org)
Izhak Haviv (izhak.haviv@bakeridi.edu.au)
Stephen B Fox (stephen.fox@petermac.org)

Version: 2 Date: 21 November 2014

Author's response to reviews:

The following changes have been made on the manuscript as requested:
1) Insertion of line numbers and page numbers
2) Insertion of acknowledgements section on p.17